Author's response to reviews

Title: Age-related changes in P wave morphology in healthy subjects

Authors:

Rasmus Havmoller (Rasmus.Havmoller@med.lu.se)
Jonas Carlson (Jonas.Carlson@med.lu.se)
Fredrik Holmqvist (Fredrik.Holmqvist@med.lu.se)
Alberto Herreros (albher@dali.eis.uva.es)
Carl J Meurling (Carl.Meurling@skane.se)
Bertil Olsson (Bertil.Olsson@med.lu.se)
Pyotr Platonov (Pyotr.Platonov@med.lu.se)

Version: 3 Date: 21 July 2007

Author's response to reviews:

Dear Sir/Madam!

We are very pleased to be able to publish our manuscript in BMC Cardiovascular Disorders and appreciate the swiftness of the review process. Below is a list of the requested formatting changes and how they have been adressed.

1. Author list: The names of the authors have been revised and are now identical in the manuscript and in the submission system.

2. Section headings: The headings and subheadings have been revised and are now in sentence case.

3. Website addresses: There are two website addresses in the manuscript. On page 4 the address for the Helsinki declaration has been removed. On page 5 "the mathworks.com" has been replaced by the company name and now reads "(The MathWorks, Inc., Natick, MA, USA)".

4. Table as figure files: The tables have been inserted in the manuscript as requested and their figure files deleted.

5. Table title and legend: The requested changes have been made and the table titles now appear above the tables and the legends below. In the manuscript they are now referred to as Table 1 and Table 2 instead of Tab. 1 and Tab. 2.

We hope that the format of the manuscript is now to your satisfaction but do not hesitate to contact me if necessary.

Sincerely Yours

Dr Rasmus Havmoller